Author’s response to reviews

Title: Engaging diverse underserved communities to bridge the mammography divide

Authors:

Kimberly K Engelman (kengelma@kumc.edu)
Ana Paula Cupertino (acupertino@kumc.edu)
Christine M Daley (cdaley@kumc.edu)
Trish Long (plong@kumc.edu)
Angelia Cully (acully@kumc.edu)
Matthew S Mayo (mmayo@kumc.edu)
Edward F Ellerbeck (eellerbe@kumc.edu)
Mugur V Geana (geanam@ku.edu)
Allen Greiner (agreiner@kumc.edu)

Version: 4 Date: 18 January 2011

Author’s response to reviews: see over
January 18, 2011

Dear BMC Public Health Editor,

Thank you for your most recent communication regarding our study protocol entitled “Engaging Diverse Underserved Communities to Bridge the Mammography Divide”.

Per your request, we have made the following revisions:
1. Ensured that the manuscript and submission system author lists are exact duplicates
2. Fixed the numerical ordering issue in the in-text table listings and ensured they corresponded properly with the table list and tables contained near the end of the manuscript
3. Revised the manuscript title to reflect sentence case
4. Renamed the Methods/Design section according to the suggestion provided
5. Revised one table’s footer (actually deleted the footer and incorporated the text in the table)
6. Reformatted table 5 to exclude merged cells that crow rows
7. Updated references #14 and #15 to make them complete
8. Deleted figure numbers and titles from figure images and closely cropped figures
9. Performed a final read through for typographical errors and made necessarily minor revisions.

Thank you for the opportunity to publish this protocol manuscript in BMC Public Health!

Sincerely,

Kimberly K. Engelman, Ph.D.
Associate Professor